\begin{table} %t4  
\caption{Proper motion information for CrA members and candidate members.\label{tab:pminfo}}
\par
%\centering
\par
\begin{tabular}{l c c c c c}
\hline\hline\noalign{\smallskip}
Name &  USNO-B1 & $\mu_{\alpha}\cos\delta$ (mas/yr) &  $\mu_{\delta}$ (mas/yr) & P &  Nobs \\
\hline
  CrA~205                           & 0526-0958780 &   $38\pm6$  & $-$$52\pm16$ & 9 & 4
  \\
  CrA~466                           & 0530-0881950 &   $24\pm54$ & $-$$64\pm2$  & 7 & 3
  \\
  CrA~468                           & 0529-0905310 &   $20\pm9$  & $-$$34\pm9$  & 9 & 4
  \\
  CrA~4107                          & 0532-0801755 & 	$8\pm1$  & $-$$22\pm6$  & 9 & 4
  \\
  CrA~4108                          & 0532-0800872 & 	$6\pm10$ & $-$$24\pm4$  & 9 & 4
  \\
  CrA~4109                          & 0532-0800993 &  $108\pm5$  &    $14\pm5$  & 9 & 4
  \\
  CrA~4110                          & 0530-0881945 &   $28\pm2$  & $-$$48\pm5$  & 9 & 4
  \\
  CrAPMS~3AB  		            & 0530-0882190 &   $54\pm20$ & $-$$28\pm7$  & 9 & 4\\
  $[$FP2007$]$ J190149.35-370028.6  & 0529-0905310 &   $20\pm9$  & $-$$34\pm9$  & 9 & 4
  \\
  G-1& 0530-0882197 &   $22\pm7$  & $-$$24\pm16$ & 9 & 4\\
  G-14                              & 0529-0905404 & $-$$8\pm1$  & $-$$24\pm22$ & 7 & 4
  \\
  G-102                             & 0529-0905023 &   $12\pm8$  & $-$$44\pm12$ & 9 & 4
  \\
  RX J1901.1-3648 & 0532-0800238 & $4\pm23$ & $-$$40\pm14$ & 9 & 4
  \\
  \hline
  CrA~J190151.7-371048.4            & 0528-0928314 &   $30\pm5 $ &  $-$$6\pm12$ & 9 & 4\\
  CrA~J190111.6-364532.0            & 0532-0800263 &   $10\pm12$ & $-$$26\pm8 $ & 9 & 4\\
\hline
\end{tabular}
\tablebib{Neuh\"auser et~al. (\cite{neuhauser00}); L\'opez Mart\'{\i} et~al. (\cite{lm05}); Forbrich \& Preibisch (\cite{forbrich07}); Sicilia-Aguilar et~al. (\cite{sicilia08}); Neuh\"auser \& Forbrich (\cite{neuhauser08}); this work.}
\end{table}